Author’s response to reviews

Title: Dual effect of the Valsalva maneuver on autonomic nervous system activity, intraocular pressure, Schlemm’s canal, and iridocorneal angle morphology

Authors:

Li Sun (dr.lilysun1996@gmail.com)
Wei Chen (450741814@qq.com)
Zhiqi Chen (66czq@163.com)
Yan Xiang (xyz2000832@163.com)
Jingmin Guo (GJM696@163.com)
Tian Hu (hutian.rush@foxmail.com)
Qiongfang Xu (492517989@qq.com)
Hong Zhang (dr_zhanghong@vip.163.com)
Junming Wang (eyedrwjm@163.com)

Version: 2 Date: 09 Dec 2019

Author’s response to reviews:

1. Authors Contributions

In the “Authors’ contributions” section of the Declarations, please ensure all authors listed in the manuscript are represented here, with exactly what they contributed to the production of this study and manuscript also listed. We currently note several authors are missing from this section e.g. authors ZC, YX, JG, TH and QX

Thank you, we have listed all authors in the “Authors’ contributions” section of the Declarations.

2. Role of Funders

Please note the role(s) of the funding body (bodies) in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript, which should be declared in the Funding section.

Thank you, we have noted the role of the funding body in the Funding section.
3. Clean manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Thank you, we have uploaded our manuscript as required.